Author's response to reviews

Title: A Supratentorial Primitive Neuroectodermal Tumor presenting with Intracranial Hemorrhage in a 42-year-old male: Case Presentation and Literature Review

Authors:

Evangelos K Papadopoulos MD (neuroevan@yahoo.gr)
Kostas N Fountas MD, PhD (fountas@med.uth.gr)
Alexandros G Brotis MD, PhD (alexbrodis@yahoo.com)
Konstantinos N Paterakis MD, PhD (kpaterakis@yahoo.com)

Version: 5 Date: 14 January 2013

Author's response to reviews: see over
Dear Editor,

All changes requested have taken place.

1) The references have been re-organised and a table summarising the results has been added.

2) The word "succumbed" has been substituted for, by the word "died".

3) Arrows highlight the details in the photos.

All changes have been highlighted in the text.

Sincerely,

On behalf of the authors

Alexandros G. Brotis, MD, PhD